Author’s response to reviews

Title: Mechanisms Underlying Bronchodilatory and Anti-inflammatory Activities of Polygonatum verticillatum: Augmented by Isolated Molecules

Authors:

Haroon Khan (hkdr2006@gmail.com)
Muhammad Saeed (saeedrph2000@yahoo.com)
Malik Hassan Mehmood (hassan.mehmood@aku.edu)
Najeeb-ur-Rehman (n_rehman5@hotmail.com)
Naveed Muhammad (drnaveedrph@gmail.com)
Ikram-ul-Haq (ikram_pharmacist@yahoo.com)
Nadeem Ashraf (nadeem11ashraf@gmail.com)
Kamal Eldin H El-Tahir (ktaher@ksu.edu.sa)
Anwarul H Gilani (anwar.gilani@aku.edu)

Version: 2 Date: 31 August 2012

Author’s response to reviews: see over
Dear Editor-in-Chief

“BMC Complementary and Alternative Medicine”

I am pleased to submit our manuscript “Mechanisms Underlying Bronchodilatory and Anti-inflammatory Activities of Polygonatum verticillatum: Augmented by Isolated Molecules” for your kind consideration in “BMC Complementary and Alternative Medicine”. We followed the instructions to author, while preparing this manuscript. The principal author and all co-authors hereby transfers, assigns and conveys all interest in and ownership of copyright until such time and unless said manuscript is rejected in writing by the Journal. We would like to state that due care has been taken to ensure the integrity of our work and the scientific reputation and that there are no financial or contractual agreements or obligations linked to the paper that might cause conflicts of interest.

We look forward to hearing your decision on this manuscript.

With kind regards,

Haroon Khan, PhD
Associate Professor and Principal, Gandhara College of Pharmacy, Gandhara University, Peshawar, KPK, Pakistan.
Tel: (+92) 91-5700033
E. mail: hkdr2006@gmail.com